\documentclass{article}

\usepackage{url}
\usepackage{supertabular}

\pagestyle{empty}

\usepackage{fullpage}

\setlength{\textwidth}{6.5in}
\setlength{\textheight}{9in}

\setlength{\oddsidemargin}{0in}
\setlength{\evensidemargin}{0in}
\setlength{\topmargin}{0in}

\newenvironment{itemize*}%
  {\begin{itemize}%
    \setlength{\itemsep}{0pt}%
    \setlength{\parskip}{0pt}}%
  {\end{itemize}}

\begin{document}

\section*{Homework Assignment 7 (due 12 noon, Fri Oct 28$^{\rm th}$)}

This assignment is composed of the following questions from Chapter 9
of \emph{An Introduction to Modern Stellar Astrophysics}.

\begin{itemize}
\item Q9.1
\item Q9.2
\item Q9.3
\item Q9.4
\item Q9.5
\item Q9.6
\item Q9.7
\end{itemize}

\end{document}
